Author's response to reviews

**Title:** Utility of clinical parameters to identify HIV infection in infants below ten weeks of age in South Africa: a prospective cohort study

**Authors:**

Heather B Jaspan (heather.jaspan@sbri.org)
Landon Myer (landon.myer@uct.ac.za)
Shabir A Madhi (ShabirM@nicd.ac.za)
Avy Violari (violari@mweb.co.za)
Diana M Gibb (dmg@ctu.mrc.ac.uk)
Wendy S Stevens (wendy.stevens@nhls.ac.za)
Els Dobbels (mcot@sun.ac.za)
Mark F Cotton (mcot@sun.ac.za)

**Version:** 4  **Date:** 3 November 2011

**Author's response to reviews:** see over
November 2, 2011

To the Editor,

Thank you for accepting our manuscript entitled “Utility of clinical parameters to identify HIV infection in infants below ten weeks of age in South Africa”. We have reviewed the formatting, and corrected headings, tables and references, as well as moved private funding to the competing interests section. We think the manuscript now adheres to the journal formatting requirements.

Sincerely,

Mark F Cotton, FCPaed (SA), PhD